Author’s response to reviews

Title: APOB codon 4311 polymorphism is associated with hepatitis C virus infection through altered lipid metabolism

Authors:

Rie Harada (lee_etan@hotmail.com)
Masako Kimura (mkaji89@gmail.com)
Yasushi Sato (yasushis2012@gmail.com)
Tatsuya Taniguchi (qyrwx456@yahoo.co.jp)
Tetsu Tomonari (tomonari.tetsu@tokushima-u.ac.jp)
Takahiro Tanaka (takahiro_tanaka2002@yahoo.co.jp)
Hironori Tanaka (hironori.t.0730@gmail.com)
Naoki Muguruma (muguruma.naoki@tokushima-u.ac.jp)
Hirohiko Shinomiya (todon@md.pikara.ne.jp)
Hirohito Honda (honda@toku-souken.jp.org)
Issei Imoto (issehgen@tokushima-u.ac.jp)
Masahiro Sogabe (sogabe.masahiro@tokushima-u.ac.jp)
Toshiya Okahisa (okahisa5505@tokushima-u-ac.jp)
Tetsuji Takayama (takayama@tokushima-u.ac.jp)

Version: 2 Date: 16 Jan 2018

Author’s response to reviews:

Marta Gritti
Editor
BMC Gastroenterology
Dear Marta Gritti,

Thank you very much for reviewing our Manuscript entitled " APOB codon 4311 polymorphism is associated with hepatitis C virus infection through altered lipid metabolism" (Manuscript ID BMGE-D-17-00542R1).

We have carried out some editorial revisions as you requested. I hope this revised manuscript is satisfactory for publication and meets with your journal policy. Again, thank you so much for your generous advice and comments on our paper.

Sincerely,

Tetsuji Takayama
Professor, Chairman, Department of Gastroenterology and Oncology,
Tokushima University Graduate School of Biomedical Sciences,
3-18-15, Kuramoto-cho, Tokushima, 770-8503,
Japan.
Tel: +81-88-633-7122
Fax: +81-88-633-9235
E-mail: takayama@tokushima-u.ac.jp